

# ARTICLE OPEN

Check for updates

# Baseline characteristics of eyes with early residual fluid post loading phase of aflibercept therapy in neovascular AMD: PRECISE study report 3

Shruti Chandra <sup>1,2</sup>, Sarega Gurudas<sup>2</sup>, Ian Pearce<sup>3</sup>, Martin Mckibbin <sup>4</sup>, Ajay Kotagiri<sup>5</sup>, Geeta Menon<sup>6</sup>, Benjamin J. L. Burton <sup>7</sup>, James Talks<sup>8</sup>, Anna Grabowska <sup>9</sup>, Faruque Ghanchi <sup>10</sup>, Richard Gale<sup>11</sup>, Andrea Giani<sup>12</sup>, Victor Chong <sup>2</sup>, Ching Ning Taffeta Chen<sup>12</sup>, Luke Nicholson<sup>1</sup>, Sridevi Thottarath<sup>1</sup>, Swati Chandak <sup>1</sup> and Sobha Sivaprasad<sup>1,2</sup>

© The Author(s) 2023

**PURPOSE:** To compare the baseline characteristics in patients with and without early residual fluid (ERF) after aflibercept loading phase (LP) in patients with treatment naïve neovascular age related macular degeneration (nAMD).

**METHODS:** Patients with nAMD initiated on LP of three intravitreal aflibercept doses were recruited from December 2019 to August 2021. Baseline demographic and OCT features associated with any ERF were analysed using Generalised Estimating Equations to account for inter-eye correlation. Receiver operating characteristic (ROC) curve was performed for selection of CST threshold.

**RESULTS:** Of 2128 patients enrolled, 1999 eyes of 1862 patients with complete data were included. After LP, ERF was present in 1000 (50.0%), eSRF in 746(37.3%) and eIRF in 428 (21.4%) eyes. In multivariable analysis of baseline features, eyes with increased central subfield thickness (CST) (OR 1.31 per 100 microns increase [95% CI 1.22 to 1.41]; P < 0.001), eyes with IRF and SRF at baseline (1.62 [95% CI 1.17 to 2.22]; P = 0.003), and those with SRF only (OR 2.26 [95% CI 1.59 to 3.20]; P < 0.001) relative to IRF only were determinants of ERF. CST  $\ge$  418 microns had 57% sensitivity and 58% specificity to distinguish ERF from no ERF at visit 4. **CONCLUSION:** On average, 50% of eyes have ERF after aflibercept LP. Clinically relevant baseline determinants of ERF include CST  $\ge$  418  $\mu$  and presence of only SRF. These eyes may require further monthly treatment before extending treatment intervals.

Eye (2024) 38:1301-1307; https://doi.org/10.1038/s41433-023-02886-1

#### INTRODUCTION

The standard of treatment for macular neovascularisation (MNV) secondary to neovascular age related macular degeneration (nAMD) is intravitreal injections of anti-vascular endothelial growth factor (anti-VEGF) agents [1]. Most anti-VEGF treatment regimens are initiated with a loading phase (LP) of monthly injections for 3 doses. The macular fluid status on optical coherence tomography (OCT) at the first appointment post LP is a decision point to assess early treatment response and customize future treatment regimen for each patient [2, 3].

A proportion of eyes have early residual fluid (ERF) following LP and this outcome is previously described as ERF or ERF-free. The ERF may be either sub-retinal fluid (SRF) and/or intraretinal fluid (IRF) and are termed eSRF and eIRF and their absences are abbreviated as eSRF-free and eIRF-free respectively [3]. With the recent availability of several anti-VEGF agents, it is now timely to explore the baseline disease characteristics of eyes with ERF after LP with aflibercept, the standard comparator used in recent clinical trials on newer agents for nAMD.

Aflibercept is a recombinant fusion protein that neutralizes all VEGF-A isoforms as well as VEGF-B and placental growth factor [4]. In the pivotal trials for aflibercept in nAMD, VIEW 1 and 2 studies, ERF was found in 22.8% of eyes at 12 weeks after aflibercept LP, with 16% having eSRF and 21.3% with eIRF [5]. Recently, secondary analysis of the pooled TENAYA and LUCERNE data showed that 33% of the patients in the aflibercept arm had ERF at 12 weeks [6]. In contrast, ERF was present in 23% of the pooled faricimab cohort at 12 weeks and predominantly driven by higher resolution of SRF [7]. None of these studies report the ERF 8 weeks after LP except the HAWK and HARRIER studies, where disease activity after LP of brolucizumab was observed in 24% of patients at 16 weeks post first injection of brolucizumab compared to 34.5% in the aflibercept arm [8].

Received: 11 August 2023 Revised: 20 November 2023 Accepted: 30 November 2023 Published online: 15 December 2023

<sup>&</sup>lt;sup>1</sup>National Institute of Health Research Moorfields Biomedical Research Centre, Moorfields Eye Hospital, London, UK. <sup>2</sup>Institute of Ophthalmology, University College, London, UK. <sup>3</sup>The Royal Liverpool and Broadgreen University Hospitals NHS Foundation Trust, Liverpool, UK. <sup>4</sup>Leeds Teaching Hospitals NHS Trust, Leeds, UK. <sup>5</sup>South Tyneside and Sunderland NHS Foundation Trust, Sunderland, UK. <sup>6</sup>Frimley Health NHS Foundation Trust, Surrey, UK. <sup>7</sup>James Paget University Hospitals NHS Foundation Trust, Norfolk, UK. <sup>8</sup>Newcastle Hospitals NHS Foundation Trust, Newcastle, UK. <sup>9</sup>King's College Hospital NHS Foundation Trust, London, UK. <sup>10</sup>Bradford Teaching Hospitals NHS Foundation Trust, Bradford, UK. <sup>11</sup>Hull York Medical School and York, University of York and Scarborough Teaching Hospital NHS Foundation Trust, York, UK. <sup>12</sup>Boehringer Ingelheim, Binger Str. 173, 55216 Ingelheim am, Rhein, Germany.

1302

These clinical trial results indicate that although most patients with new nAMD respond rapidly to anti-VEGF treatment, ERF is seen in a proportion of patients irrespective of the agents used. These patients would likely require more aggressive treatment compared to ERF-free eyes. As clinical trials are restricted by their eligibility criteria, the proportion of patients with ERF in clinical practice and the baseline demographic and imaging factors associated with ERF is unclear.

The aim of this study was to report the differences in baseline characteristics in patients with ERF versus ERF free after aflibercept LP in patients with treatment naïve nAMD in clinical practice.

# METHODS

The PRECISE study was conducted over 10 National Health Service retinal centres in the United Kingdom (UK). The aim of the study was to conduct indepth analysis of the response of treatment naïve nAMD to LP of aflibercept therapy. The study was approved by the Institutional Review Board of National Research Service (REC number 19/LO/1385) and followed the tenets of the Declaration of Helsinki. Written informed consent was obtained from all participants before study entry (ISRCTN28276860). Recruitment period was from 18/12/2019 to 04/08/2021.

#### Patient eligibility criteria

Consent to the study was obtained from patients aged 50–100 years who were initiated on intravitreal aflibercept after diagnosis of nAMD by the investigators at the local sites. Both visual acuity (VA) and Spectralis OCT scans at baseline and within 10 weeks from 3rd loading dose injection were mandatory inclusion criteria. Both eyes of an individual were recruited if eligible. Exclusion Criteria included co-existent ocular disease that, in the opinion of the investigator, might affect or alter visual acuity during the study, poor image quality and missing scans.

#### Data collection

Demographic data collected included age, gender and ethnicity. Eye level data included VA and OCT scans done at baseline and within 10 weeks after 3rd injection. Anonymised data were entered in a web-based database (Playon Ltd, Bengaluru, India). Routine raster scan protocol centred at the fovea (range  $20^{\circ}x25^{\circ}$  to  $30^{\circ}x20^{\circ}$ ) in the Heidelberg Engineering Systems "Spectralis HRA + OCT" or "Spectralis OCT" systems. The line scan protocols on Heidelberg Spectralis scans ranged from 19 to 49 (19, 25, 31 or 49). Retinal images were anonymised and transferred by encrypted USB to Moorfields Eye Hospital for grading.

#### OCT grading protocol

A pre-defined OCT grading manual was used to grade the OCT scans. The images were graded by five Medical Retina Fellows after grading on a test set of 50 OCT images. Each scan was corrected for any segmentation errors and foveal centration. The grading results were recorded on standardised case report forms. The ERF was defined as presence of any elRF or eSRF anywhere in the 6×6 mm macula scan at the first clinic appointment after LP. The final visit was up to 140 days after the first injection were permitted. The CST was measured from the internal limiting membrane (ILM) to the Bruch's membrane (BrM). The type of neovascularization was graded on OCT as per the Consensus on Neovascular Age-Related Macular Degeneration Nomenclature Study Group (CONAN) classification [9].

*Outcomes.* The primary outcome of the study was the presence of ERF at the final visit. Other outcomes included the presence of eSRF or eIRF at the final visit.

#### Statistical analysis

Data were summarised with mean  $\pm$  SD or median (interquartile range [IQR]) for normally and non-normally distributed continuous variables, respectively and *n* (%) for categorical variables. Univariate and multivariable associations between demography and OCT features and the binary outcomes for ERF, eSFF and eIRF were reported using Odds Ratio (95% CI) and *P*-value. Generalised Estimating Equations (GEE) with an exchangeable working correlation structure were used to account for the

within-participant correlation among those with data from both eyes [10, 11]. Black, south Asians, other Asians and other ethnicities were categorised as Non-White in models due to insufficient sample size. CST was analysed as a continuous variable and in guartiles, ranging from low CST (quartile 1) to high CST (quartile 4). Receiver operating characteristic (ROC) curve analysis was performed for CST and IRF ± SRF using clustered bootstrap with 1000 replicates to estimate the bias corrected 95% confidence intervals for the area under the curve (AUC), and optimal thresholds for CST were selected based on maximising Youden's index, defined as sensitivity + specificity - 1 [12, 13]. Bootstrapping allows for probability-based inference for the AUC and corrects for the inter-eye correlation with respect to the estimation of the standard error of the estimated AUC. Sensitivity analysis was performed to assess associations of MNV features with the study outcomes, excluding patients that attended their final visit less than 104 days from their first aflibercept injection. Statistical analysis was undertaken using R version 4.1.2 statistical software package and Stata MP 15.

## RESULTS

A total of 2039 gradable eyes of 1901 patients were initiated on aflibercept therapy for nAMD. The flow chart shows how 1999 eyes of 1862 patients was derived to form the sample for analysis (Fig. S1).

Table 1 shows the age wise distribution, vision and some of the imaging characteristics of the eyes included in the study. Table S1 shows the detailed demographic, clinical and ocular characteristics. Mean age was 79.3 (SD 7.8) years, 1126 (60.5%) were women and 1772 (95.2%) participants were white. The median time interval between the first aflibercept injection and the final visit was 112 days (IQR 98 to 119 days) in the total cohort, and modes located at 86.7 (~12 weeks) and 114 days (~16 weeks). CST was categorized as less than or equal to 340  $\mu$ m in quartile 1 (Q1), 341  $\mu$ m to 415  $\mu$ m in quartile 2 (Q2), 416  $\mu$ m to 525  $\mu$ m in quartile 3 (Q3) and greater than 525  $\mu$ m in quartile 4 (Q4). Median CST was 416  $\mu$ m (IQR 340  $\mu$ m to 526  $\mu$ m).

# Baseline macular fluid status

At baseline, a third of eyes (672 of 1999 [33.6%]) presented with both IRF and SRF. More than 80% had SRF (1654 of 1999 [82.7%]) and approximately half of the eyes had IRF (1017 of 1999 [50.9%]). A total of 982 (49.1%) had SRF without IRF at presentation.

#### Early residual fluid (ERF)

After LP, ERF was present in 1000 eyes (50.0%, Fig. 1). The proportion with eSRF and eIRF were 37.3% (N = 746) and 21.4% (N = 428) respectively. The proportion of eyes with co-existent eSRF and eIRF was 8.7% (N = 174 eyes), and 50.0% of total cohort had no macular fluid (N = 999 eyes). Table 2 shows the baseline and post LP VA (unadjusted and adjusted) based on the distribution of ERF. Eyes with residual eIRF had the worst VA across all groups.

#### Associations of ERF and its distribution

Univariate and multivariable analysis of participant demographic and OCT characteristics were performed to study its associations with presence of ERF, eSRF and eIRF (Table S2 and Fig. 2).

Variables with increased odds of ERF in multivariable analysis included; presence of IRF and SRF, SRF only and increased CST. Variables associated with a reduction of odds of ERF included participants aged 80 years and above, eyes with retinal angiomatous proliferation (RAP) relative to Type 1, presence of atrophy or outer retinal tubulation, and ungradable ellipsoid zone (EZ) and external limiting membrane (ELM).

Non-whites, increased CST, presence of IRF and SRF and presence of SRF only were associated with increased odds of eSRF. While visual acuity <54 ETDRS letters, RAP lesions, presence of atrophy or outer retinal tubulation were associated with reduced odds of SRF at visit 4.

Variable	Overall, N = 1999	Visit 4 <sup>a</sup>					
	eyes or 1802 patients	ERF status		eSRF status		elRF status	
		No ERF, N = 999 eyes of 950 patients	ERF, N = 1000 eyes of 973 patients	No eSRF, N = 1253 eyes of 1174 patients	eSRF, N = 746 eyes of 734 patients	No eIRF, N = 1571 eyes of 1481 patients	eIRF, N = 428 eyes of 417 patients
Patient level (N= 1862) Age, years	79.3 (7.8)	80.4 (7.7)	78.3 (7.6)	80.4(7.7)	77.7 (7.5)	79.3 (7.9)	79.8 (7.3)
Eye level ( $N = 1999$ eyes)							
Mean Baseline visual acuity, ETDRS letters	58.0 (14.5)	57.9 (14.3)	58.1 (14.9)	57.0 (14.5)	59.5 (14.6)	59.1 (14.1)	53.5 (15.6)
Baseline visual acuity cated	gories, ETDRS letters						
<54	648 (32.4%)	318 (31.8%)	330 (33.0%)	433 (34.6%)	215 (28.8%)	449 (28.6%)	199 (46.5%)
54-67	715 (35.8%)	373 (37.3%)	342 (34.2%)	459 (36.6%)	256 (34.3%)	573 (36.5%)	142 (33.2%)
≥68	636 (31.8%)	308 (30.8%)	328 (32.8%)	361 (28.8%)	275 (36.9%)	549 (34.9%)	87 (20.3%)
Central subfield thickness in microns	416 (340, 526)	393 (322, 492)	437 (362, 567)	401 (326, 509)	436 (364, 567)	404 (334, 510)	467 (366, 596)
Presence of Atrophy	421 (21.1%)	265 (26.5%)	156 (15.6%)	341 (27.2%)	80 (10.7%)	314 (20.0%)	107 (25.0%)
Presence of fibrosis	292 (14.6%)	128 (12.8%)	164 (16.4%)	189 (15.1%)	103 (13.8%)	183 (11.6%)	109 (25.5%)
Presence of SHRM	1164 (58.2%)	576 (57.7%)	588 (58.8%)	722 (57.6%)	442 (59.2%)	896 (57.0%)	268 (62.6%)
Presence of SDD	598 (29.9%)	337 (33.7%)	261 (26.1%)	429 (34.2%)	169 (22.7%)	466 (29.7%)	132 (30.8%)
Presence of VMT or ERM	251 (12.6%)	125 (12.5%)	126 (12.6%)	179 (14.3%)	72 (9.7%)	176 (11.2%)	75 (17.5%)
EZ and ELM combination							
Intact EZ and ELM	847 (42.4%)	378 (37.8%)	469 (46.9%)	451 (36.0%)	396 (53.1%)	736 (46.8%)	111 (25.9%)
Either EZ or ELM loss	584 (29.2%)	313 (31.3%)	271 (27.1%)	427 (34.1%)	157 (21.0%)	401 (25.5%)	183 (42.8%)
Both ungradable	568 (28.4%)	308 (30.8%)	260 (26.0%)	375 (29.9%)	193 (25.9%)	434 (27.6%)	134 (31.3%)
CONAN consensus on neovas external limiting membrane,	cular AMD nomenclature, <i>ell</i> <i>EZ</i> ellipsoid zone, <i>HRF</i> hyper	RF early intraretinal fluid, E rreflective foci, <i>IQR</i> interquertine interquertin	RF early residual fluid, ERA Lartile range, IRF intraretin	A epiretinal membrane, eSRF and fluid, MNV macular neova	early subretinal fluid, <i>ETI</i> scularization, <i>OCT</i> optica	JRS early treatment diabetic I I coherence tomography, Of	retinopathy study, <i>ELM</i> ? odds ratio, ORT outer

material, SRF subretinal fluid, VA visual acuity, VMT vitreomacular traction. <sup>3</sup>The same patients may be present in different residual macular fluid groups, hence the number of patients may not sum up to the total of 1862 patients.



1304



 Table 2.
 Presenting and post-loading VA based on distribution of ERF at visit 4.

ERF at	Baseline Mean	Post-loading (Visit 4)					
VISIT 4	VA (SD)	Mean VA (SD)	Adjusted Mean VA (SE) <sup>a</sup>				
ERF	58.1 (SD 14.8)	61.3 (16.5)	61.7 (SE 0.35)				
No ERF	57.9 (SD 14.3)	64.4 (16.0)	63.4 (SE 0.36)				
eSRF (±eIRF)	59.5 (SD 14.5)	63.2 (13.9)	62.1 (SE 0.41)				
eSRF only	61.5 (SD 13.4)	65.2 (12.8)	62.7 (SE 0.45)				
No eSRF	57.1 (SD 14.5)	62.1 (16.0)	62.8 (SE 0.32)				
eIRF (±eSRF)	53.6 (SD 15.5)	57.2 (16.4)	60.3 (SE 0.56)				
eIRF only	54.0 (SD 15.0)	57.6 (17.0)	60.4 (SE 0.71)				
No elRF	59.2 (SD 14.0)	64.0 (14.6)	63.2 (SE 0.29)				

*ERF* early residual fluid, *eSRF* early subretinal fluid, *eIRF* early intraretinal fluid. <sup>a</sup>GEE model with visit 4 visual acuity as the outcome and adjusted for baseline visual acuity.

Variables that showed increased odds for IRF at visit 4 only included increased CST and presence of vitreomacular traction or ERM. SRF only (without IRF) at baseline was associated with reduced odds of eIRF.

Sensitivity analysis excluding 592 eyes where final visit occurred less than 104 days from the first aflibercept injection showed that our results remain robust to the main analysis (Table S3).

## **Optimal threshold for CST**

ROC analysis was performed to find the optimal cut-off point for CST capable of discriminating presence of ERF, eSRF and eIRF at visit 4 (Fig. 3). AUCs were comparable and ranged from 0.59 to 0.61 for distinguishing presence and absence of ERF, eSRF and eIRF using CST alone. CST  $\geq$  418 microns had 57% sensitivity and 58% specificity to distinguish ERF from no ERF at visit 4. While CST  $\geq$  359 microns was able to distinguish eSRF from no eSRF at visit 4 with 78 and 36% sensitivity and specificity respectively. CST  $\geq$  448 microns identified eIRF with 54% sensitivity and confirmed no eIRF with 64% specificity. CST together with SRF and/or IRF (IRF only, SRF and IRF, SRF only) yielded AUCs 0.65 (95% bias-corrected CI 0.64–0.68), 0.77 (95% bias-corrected CI 0.72–0.77) for predicting eRF, eSRF and eIRF respectively.

#### DISCUSSION

In this study, we report the proportion of patients with ERF, eSRF, eIRF after aflibercept LP for nAMD and their determinants at baseline.

Fifty percent of the cohort had ERF after LP and this proportion is similar to that achieved by the aflibercept cohort at 8 weeks post loading injections in TENAYA and LUCERNE. When we consider the distribution of ERF, the proportion of patients with no IRF (with or without SRF) was 78.6% and SRF with or without IRF was 62.7% in our study. These results also mirror those obtained 8 weeks post loading in the aflibercept arm of TENAYA and LUCERNE (76% with no IRF and 62% with no SRF) [6].

Baseline factors associated with ERF included non-white ethnicity, males, increased CST, eyes presenting with IRF and SRF or with SRF only. These results indicate that patients with these features are likely to require further monthly injections for complete fluid resolution.

A key observation is that high baseline CST is associated with higher prevalence of ERF. Similar observations were noted in the VIEW 1 and 2 studies [5]. Post-hoc exploration of baseline features in the ALTAIR study also showed that patients with increased central retinal thickness, high PED height and absence of PCV or subretinal haemorrhage were more likely to have retinal fluid at week 16 [14]. Secondary analysis of the combined faricimab arms in TENAYA and LUCERNE also showed that eyes with increased CST at baseline were more likely to require 8 weekly dosing compared to the other two cohorts dosed at extended intervals [15]. Similar results were also observed in HAWK and HARRIER analysis [8]. These findings highlight that irrespective of the drug used, eyes presenting with high CST are the 'difficult to treat' group and are likely to require frequent injections. Therefore, these eyes necessitate further monthly injections before considering extension of treatment intervals. In this study, a CST of  $\geq$  418 microns is likely associated with ERF.

Although non-white ethnicity was associated with ERF and eSRF, only 5% of the study population were of non-white origin. Therefore, a further study comparing similar proportions of white and non-white patients is required to validate these findings.

Patients aged 80 years or older were less likely to have ERF. This may be partly explained by the higher prevalence of RAP and atrophy in older individuals, which were both also associated with decreased odds of having ERF in this study. This finding is in keeping with the secondary analysis of TENAYA and LUCERNE, which also demonstrated higher proportions of RAPs in the eyes that were maintained in the Q16W arm [15].

When we consider the relationship of VA in eyes with ERF and its distribution, the findings in our real-life study are comparable to the results of a post-hoc analysis of aflibercept LP in the ARIES study [16, 17]. Both the presenting VA and the VA outcome adjusted for baseline VA and other potential confounders were numerically higher in eyes with eSRF. The converse was true for eIRF. The negative impact of IRF has been established in previous reports at various time-points [14, 18–21]. Our results that high CST and poorer VA at baseline are associated with eIRF indicate that these eyes also need to be treated more aggressively.

Interestingly, baseline atrophy was associated with less odds of both ERF and eSRF. The complete loss of the RPE barrier may indeed facilitate unimpeded passage of aflibercept to the MNV. This may also explain why eyes with ungradable ellipsoid layer are less likely to have ERF and SRF post-loading. In active MNV eyes, features like subretinal fluid or haemorrhage or subretinal hyperreflective material make it challenging to grade ellipsoid layer due to significant back shadowing or blurring of outer layer details. As such any underlying EZ/ELM loss indicating atrophy might be missed at baseline. On the contrary, baseline atrophy did not influence the resolution of IRF.

			ERF			eSRF			elRF	
Age, years <sup>a</sup>	70-79		-			-				
	>=80	н	•		-	•		•	+	
Gender <sup>b</sup>	Male		•			•			•	
Ethnicityc	Non-white									
MNVd	Non-foveal				-	<b>-</b>			•	
MNV Type <sup>e</sup>	Type 2 +mixed	1	•			•				
	RAP	н	•			H			-+-	
	PCV	-	<b>+</b> · ·		-	•				
MNV complex <sup>f</sup>	Yes		•					•	•	
Visit 1 VA <sup>g</sup>	54-67		•			•			<b>---</b>	
	<54		•		-	•			<b>•</b> •	
CST	CST, per 100µm		•			•			•	
IRF+/-SRF <sup>h</sup>	IRF and SRF					-	•		•••	
	SRF only							• <b>•</b> •		
PED <sup>f</sup>	Non-foveal		+		i I I	<b></b>				
	Foveal involving		<b>+</b> -							
Atrophy or ORT <sup>f</sup>	Yes	-						-	•	
Fibrosis <sup>f</sup>	Yes		•							
SHRMf	Yes		•			•			-	
Drusen <sup>f</sup>	Yes		· • ·			++			• <b>•</b> •	
SDDf	Yes		• <b>•</b> •			•			+ <b>+</b> +	
HRFf	Yes		•			•			•	
VMT or ERM <sup>f</sup>	Yes		•••		-	•				
EZ/ELMi	EZ/ELM loss	-	•		-	•				
	Both ungradable	H				•				
		0.3	1.0 3.0	10.0 30.0	0.3	1.0 3.0 OP	10.0 30.0	0.3	1.0 3.0	10.0 30.0

Fig. 2 Plot of Odds Ratio with 95% Cls showing baseline demographic and OCT characteristics associated with eRF, eSRF and elRF at visit 4 – multivariable analysis using Generalised Estimating Equations (GEE). Non-White includes Black, South Asian, other Asian or other ethnic categories, Cl confidence interval, CONAN consensus on neovascular AMD nomenclature, elRF early intraretinal fluid, eRF early residual fluid, ERM epiretinal membrane, eSRF early subretinal fluid, ETDRS early treatment diabetic retinopathy study, ELM external limiting membrane, EZ ellipsoid zone, GEE generalized estimating equation, HRF hyperreflective foci, IQR interquartile range, IRF intraretinal fluid, MNV macular neovascularization, OCT optical coherence tomography, OR Odds Ratio, ORT outer retinal tubulation, PCV polypoidal vasculopathy, PED pigment epithelial detachment, RAP retinal angiomatous proliferation, SD standard deviation, SDD subretinal drusenoid deposits, SHRM subretinal hyperreflective material, SRF subretinal fluid, VA visual acuity, VMT vitreomacular traction. The ratio axis is displayed on the logarithmic scale to provide a visual description of the uncertainty associated with each estimate. Reference categories were: a<70 years, bFemale, cWhite, dMNV foveal involving, eType 1, fAbsent or No,  $gVA \ge 68$  ETDRS letters, hIRF only, iEZ and ELM intact.

1305



**Fig. 3** The receiver operating characteristic (ROC) curves for CST in detecting eRF, eSRF and eIRF at visit 4. CST central subfield thickness, AUC area under the curve, TPR true positive rate (Sensitivity), TNR true negative rate (Specificity), eRF early residual fluid, eSRF early subretinal fluid, eIRF early intraretinal fluid. Optimal threshold points that maximize Youden's index indicated in red. AUC and bias corrected 95% confidence intervals for CST were 0.61 (95% CI 0.58–0.63) for eRF, 0.59 (95% CI 0.56–0.62) for eSRF and 0.61 (95% CI 0.58–0.64) for eIRF. AUCs and bias corrected 95% confidence intervals for CST and IRF ± SRF were 0.65 (95% bias-corrected CI 0.64–0.68) for eRF, 0.77 (95% bias-corrected CI 0.75–0.79) for eSRF and 0.74 (95% bias-corrected CI 0.72–0.77) for eIRF.

Our study has the following strengths. This study analyzed a large real-world data of nearly 2000 eyes collected from 10 centers in UK, thereby representative of clinical practices across various centers. In addition, the data is uniform with all receiving three loading doses of aflibercept for treatment naïve neovascular AMD eyes and all imaging was on Heidelberg, Spectralis SD-OCT. All eyes underwent exhaustive and meticulous grading after manual correction of segmentation, if required. Presence of specific imaging biomarkers that could potentially determine residual macular fluid, including but not limited to SHRM, ORT, VMT, ERM, loss of ELM and EZ were graded by trained graders with excellent inter-grader agreement. As both eyes of some patients were included in the study robust analytical tools like generalized estimating equations were employed to account for any effect. The analyses were also adjusted for VA and time to follow-up to ensure strength of results. Finally, this study provides unique data on baseline determinants of ERF in real-life. However, there are some limitations to this study. The classification of MNV subtypes is OCT based CONAN classification and we did not have concurrent fluorescein or indocyanine green angiography to confirm the MNV subtype [9].

In conclusion, ERF is present in 50% of individuals treated with aflibercept LP. The baseline characteristics, including non-white ethnicity, males, increased CST (418 microns or more), eyes presenting with IRF and SRF or with SRF are at risk of developing ERF. These eyes are likely to require more monthly injections to attain stability before extension of treatment intervals are planned.

#### SUMMARY

What was known before

 Aflibercept is the most commonly used first-line agent for treatment of naive neovascular AMD (nAMD) in the UK. Aflibercept is also the standard comparator used in recent clinical trials on newer agents for nAMD. A proportion of eyes have early residual fluid (ERF) following LP and this outcome is previously described as ERF or ERF-free. As clinical trials are restricted by their eligibility criteria, the proportion of patients with ERF in clinical practice and the baseline demographic and imaging factors associated with ERF is unclear.

#### What this study adds

• Fifty percent of the cohort had ERF after LP in the real world setting. This proportion is similar to that achieved by the

aflibercept cohort across various clinical trials. Baseline factors associated with ERF included non-white ethnicity, males, increased CST, eyes presenting with IRF and SRF or with SRF only. These results indicate that patients with these features are likely to require further monthly injections for complete fluid resolution. Key observation is eyes with CST > 418 and SRF only at baseline are clinically relevant markers that determine ERF post loading.

#### DATA AVAILABILITY

The data that support the findings of this study are not openly available due to reasons of sensitivity and are available from the corresponding author upon reasonable request.

#### REFERENCES

- Gemenetzi M, Patel PJ. A systematic review of the treat and extend treatment regimen with anti-VEGF agents for neovascular age-related macular degeneration. Ophthalmol Ther. 2017;6:79–92.
- Amoaku WM, Chakravarthy U, Gale R, Gavin M, Ghanchi F, Gibson J, et al. Defining response to anti-VEGF therapies in neovascular AMD. Eye 2015;29: 721–31. https://doi.org/10.1038/eye.2015.48
- Wu AK, Perkins SW, Sachin S, Singh RP. The impact of early residual fluid after anti-vascular endothelial growth factor initiation in patients with neovascular age-related macular degeneration: a meta-analysis review. Ophthalmic Surg Lasers Imaging Retin. 2022;53:506–13. https://doi.org/10.3928/23258160-20220726-01
- Heier JS, Brown DM, Chong V, Korobelnik JF, Kaiser PK, Nguyen QD. et al. Intravitreal aflibercept (VEGF trap-eye) in wet age-related macular degeneration. Ophthalmology. 2012;119:2537–48. https://doi.org/10.1016/j.ophtha. 2012.09.006
- Waldstein SM, Simader C, Staurenghi G, Chong NV, Mitchell P, Jaffe GJ, Lu C, Katz TA. Morphology and visual acuity in aflibercept and ranibizumab therapy for neovascular age-related macular degeneration in the view trials. Ophthalmology 2016;123:1521–9. https://doi.org/10.1016/j.ophtha.2016.03.03
- Heier JS, Khanani AM, Quezada Ruiz C, Basu K, Ferrone PJ, Brittain C. et al. Efficacy, durability, and safety of intravitreal faricimab up to every 16 weeks for neovascular age-related macular degeneration (TENAYA and LUCERNE): two randomised, double-masked, phase 3, non-inferiority trials. Lancet. 2022;399:729–40. https://doi.org/10.1016/S0140-6736(22)00010-1
- Cheung CMG, Guymer RH, Demetriades AM, Margaron P, Ruiz CQ, Silverman D, et al. Faricimab in neovascular age-related macular degeneration (nAMD): efficacy, safety, and durability through week 48 in the Phase 3 TENAYA and LUCERNE Trials [Conference presentation]. 22nd EURETINA 2022, Hamburg, Germany. 2023. https://euretina.org/resource/abstract\_2022\_faricimab-inneovascular-age-related-macular-degeneration-namd-efficacy-safety-anddurability-through-week-48-in-the-phase-3-tenaya-and-lucerne-trials/
- Dugel PU, Koh A, Ogura Y, Jaffe GJ, Schmidt-Erfurth U, Brown DM, et al. HAWK and HARRIER: Phase 3, multicenter, randomized, double-masked trials of brolucizumab for neovascular age-related macular degeneration. Ophthalmology. 2020;127:72–84. https://doi.org/10.1016/j.ophtha.2019.04.017

- Spaide RF, Jaffe GJ, Sarraf D, Freund KB, Sadda SR, Staurenghi G. et al. Consensus nomenclature for reporting neovascular age-related macular degeneration data: consensus on neovascular age-related macular degeneration nomenclature study group. Ophthalmology. 2020;127:616–36. https://doi.org/10.1016/j.ophtha. 2019.11.004
- Hardin J, Hilbe J. Generalized estimating equations. 2nd ed., New York: Chapman and Hall/CRC; 2013. https://doi.org/10.1201/b13880
- 11. Liang KY, Zeger SL. Longitudinal data analysis using generalized linear models. Biometrika. 1986;73:13-22.
- Ying GS, Maguire MG, Glynn RJ, Rosner B. Tutorial on biostatistics: receiveroperating characteristic (ROC) analysis for correlated eye data. Ophthalmic Epidemiol. 2022;29:117–27. https://doi.org/10.1080/09286586.2021.1921226
- Huang FL. Using cluster bootstrapping to analyze nested data with a few clusters. Educ Psychol Meas. 2018;78:297–318. https://doi.org/10.1177/ 0013164416678980
- 14. Ohji M, Okada AA, Sasaki K, Moon SC, Machewitz T, Takahashi K. et al. Relationship between retinal fluid and visual acuity in patients with exudative agerelated macular degeneration treated with intravitreal aflibercept using a treatand-extend regimen: subgroup and post-hoc analyses from the ALTAIR study. Graefes Arch Clin Exp Ophthalmol. 2021;259:3637–47. https://doi.org/10.1007/ s00417-021-05293-y
- 15. Lai TYY, Michels S, Demetriades AM, Ruiz CQ, Silverman D, Ives J, et al. Faricimab in neovascular age-related macular degeneration: 48-week results by dosing interval cohort in the phase 3 TENAYA and LUCERNE Trials[Conference presentation]. 22nd EURETINA 2022, Hamburg, Germany. 2023. https://euretina.org/ resource/abstract\_2022\_faricimab-in-neovascular-age-related-maculardegeneration-48-week-results-by-dosing-interval-cohort-in-the-phase-3-tenayaand-lucerne-trials/
- Chaudhary V, Holz FG, Wolf S, Midena E, Souied EH, Allmeier H, et al. Association between visual acuity and fluid compartments with treat-and-extend intravitreal aflibercept in neovascular age-related macular degeneration: an ARIES post hoc analysis. Ophthalmol Ther. 2022;11:1119–30. https://doi.org/10.1007/s40123-022-00491-1
- Chaudhary V, Matonti F, Zarranz-Ventura J, Stewart MW. Impact of fluid compartments on functional outcomes for patients with neovascular age-related macular degeneration: a systematic literature review. Retina. 2022;42:589–606. https://doi.org/10.1097/IAE.00000000003283
- Waldstein SM, Wright J, Warburton J, Margaron P, Simader C, Schmidt-Erfurth U. Predictive value of retinal morphology for visual acuity outcomes of different ranibizumab treatment regimens for neovascular AMD. Ophthalmology. 2016;123:60–69. https://doi.org/10.1016/j.ophtha.2015.09.013
- Guymer RH, Markey CM, McAllister IL, Gillies MC, Hunyor AP, Arnold JJ, et al. Tolerating subretinal fluid in neovascular age-related macular degeneration treated with ranibizumab using a treat-and-extend regimen: FLUID study 24month results. Ophthalmology. 2019;126:723–34. https://doi.org/10.1016/ j.ophtha.2018.11.025
- Sharma S, Toth CA, Daniel E, Grunwald JE, Maguire MG, Ying GS, et al. Macular morphology and visual acuity in the second year of the comparison of age-related macular degeneration treatments trials. Ophthalmology. 2016;123:865–75. https:// doi.org/10.1016/j.ophtha.2015.12.002
- Sadda S, Holekamp NM, Sarraf D, Ebraheem A, Fan W, Hill L, et al. Relationship between retinal fluid characteristics and vision in neovascular age-related macular degeneration: HARBOR post hoc analysis. Graefes Arch Clin Exp Ophthalmol. 2022;260:3781–9. https://doi.org/10.1007/s00417-022-05716-4

## **AUTHOR CONTRIBUTIONS**

Conceptualization: ShC and SS; Data curation: ShC, ST and SC; Formal analysis: ShC, SG and SS; Funding acquisition: AnG, VC and SS; Investigation: ShC, SG and SS; Methodology: ShC, SG and SS Project administration: AnG and SS; Resources: ShC, GM, BJB, IP, MM, ST, SC, AK, JT, AG, FG, RG, LK and SS; Supervision: AnG, VC and SS; Visualization: SG and ShC; Writing – original draft: ShC, SG and SS; Writing - review & editing: AnG, VC and SS Review and approval of final manuscript: ShC, SG, AnG, VC. SS, GM, BJB, IP, MM, ST,SC, AK, JT, AG, CC, FG, RG and LN.

FUNDING

The investigator-initiated study was funded by Boehringer Ingelheim International GmbH (Grant number: SIVS1045). This study was also supported by the NIHR Biomedical Research Centre and Clinical Research Facility at the Moorfields Eye Hospital National Health Service Foundation Trust and the University College London Institute of Ophthalmology. Boehringer Ingelheim was involved in the review and approval of the manuscript.

#### **COMPETING INTERESTS**

SSi reported receiving financial support from AbbVie, Apellis, Bayer, Biogen, Boehringer Ingelheim, Novartis, Eyebiotech, Iveric Bio, Janssen Pharmaceuticals, Nova Nordisk, Optos, Ocular Therapeutix, OcuTerra, Roche. Victor Chong is an employee of Janssen R&D and previously of Boehringer Ingelheim. Andrea Giani is an employee of Boehringer Ingelheim. Chingning, Taffeta, Chen is an employee of Boehringer Ingelheim. Benjamin Burton has been on advisory boards and received international conference attendance sponsored by Novartis, Roche and Baver, Geeta Menon has conducted consultancy-advisory boards for Novartis, Bayer and Allergan, received educational travel grants from Novartis, Bayer, Allergan. Ian Pearce has received lecture fees from Allergan, Bayer, Roche, Heidelberg and Novartis, consultancy fees from Allergan, Alimera, Bayer and Novartis and travel fees from Allergan, Bayer and Novartis. Faruque Ghanchi has received honorarium for consultancy-advisory boards from Alimera, Allergan, Bayer, Novartis, Oxford BioElectronics, Roche, educational travel grants from Allergan, Bayer, Novartis. Martin McKibbin has received lecture and advisory board honoraria from Bayer and Novartis and an educational travel grant from Bayer. Richard Gale has conducted consultancy-advisory boards for Novartis, Bayer, Biogen, Boehringer Ingelheim and Allergan, Alimera, Santen, received educational travel grants from Novartis, Bayer, Allergan, Heidelberg Engineering, James Talks is a consultant for Bayer and Roche, received grant support from Bayer, Novartis and Heidelberg Engineering, and is involved in research for Allergan, Alimera, Roche, Bayer, Novartis and Boehringer-Ingelheim. AK received travel support from Novartis, Bayer, and Allergan, and speaker fees from Allergan and Bayer. LN received advisory boards honoraria from Roche, Abbvie and Bayer. SS, SC, and FG. Are on the Editorial board for Eye.

#### **ADDITIONAL INFORMATION**

**Supplementary information** The online version contains supplementary material available at https://doi.org/10.1038/s41433-023-02886-1.

**Correspondence** and requests for materials should be addressed to Sobha Sivaprasad.

Reprints and permission information is available at http://www.nature.com/ reprints

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

© The Author(s) 2023